Author’s response to reviews

Title: Inhibition of Airway Remodeling and Inflammatory Response by Icariin in Asthma

Authors:

Lingli Hu (lenleho@163.com)
Lulu Li ( lululeehui@163.com)
Hongying Zhang (zhy3049@163.com)
Qiuping Li (liqiuping0917@126.com)
Shan Jiang (js1124@126.com)
Jian Qiu (qiujian918@163.com)
Jing Sun (sjing0610@163.com)
Jingcheng Dong (jcdong2004@126.com)

Version: 4 Date: 31 Oct 2019

Author’s response to reviews:

We very much appreciate the editor’s comments concerning our manuscript. These comments are all valuable and very helpful for revising and improving our paper. We have studied comments carefully and have made correction. In response to these comments we offer a point by point rebuttal as detailed below:

Editor Comments:

Comment 1. We note that the current submission contains some textual overlap with other previously published works, in particular:
This overlap exists throughout your manuscript.
While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work. Please be informed that we cannot proceed with handling your manuscript before this issue is resolved, and the sections of text in question have been reformulated.
Response: We are very sorry for the textual overlap of this manuscript and previous published articles. https://www.frontiersin.org/articles/10.3389/fimmu.2016.00696/full is the article published by author (Lingli Hu) in 2016. According to the editor's suggestion, we have minimized these overlap.
Comment 2. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.
Response: Thank you for this comment. The funding body did not have any role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. The above statements have been added in funding section.

Comment 3. Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Response: We appreciate for your warm work and reminders earnestly. A single, final, clean manuscript has been uploaded, and responses to the reviewers'/editors’ comments have been put in the Response to Reviewers box in Editorial Manager.

We now resubmit our revised manuscript as a letter to the editor according to editor’s comments.